Author's response to reviews

**Title:** Sertraline plus deanxit to treat patients with depression and anxiety in chronic somatic diseases: a randomized controlled trial

**Authors:**

- Limin Wang (wanglimingd@126.com)
- Zhuoyuan Zhong (zhuoyuan@163.com)
- Jingyang Hu (15913192315@126.com)
- Xiaoming Rong (irisrong1227@hotmail.com)
- Jun Liu (13609794537@126.com)
- Songhua Xiao (songhuaxiao@aliyun.com)
- Zhonglin Liu (13719438236@126.com)

**Version:** 2  
**Date:** 17 October 2014

**Author's response to reviews:** see over
Dear Editors,

We are pleased to submit our manuscript entitled “Sertraline plus deanxit to treat patients with depression and anxiety in chronic somatic diseases: a randomized controlled trial” for your consideration as a research article for publication in BMC Psychiatry. Our original unpublished work is reported in this manuscript which has not been submitted elsewhere for publication and not under review of any other journals.

According to the instructions, we have made some changes for our manuscript. Firstly, the title page contains all requested items. Secondly, line and page numbers without page breaks have been applied in this manuscript. Thirdly, tables have been arranged after the figure legends and references.

All of the co-authors are in agreement with the contents of this submitted manuscript and have no conflict of interest to declare. The corresponding authors are as follows:

Name: Zhonglin Liu

E-mail address: zhonglinliu@126.com

Telephone: +8602081332621

Fax number: +8602081332833

Postal address: Department of Neurology. Sun Yat-sen Memorial
Hospital of Sun Yat-sen University, No.107, Yan Jiang West Road, Guangzhou, Guangdong, China.

We hope that this manuscript is suitable for publication in BMC Psychiatry. Please do not hesitate to contact me with any questions or concerns related to this manuscript. I look forward to hearing from you soon.

Sincerely yours,

Zhonglin Liu